

Dear Editor:

Thank you for your kind letter of March 14, 2021. We revised the manuscript in accordance with the reviewer's comments, and carefully proof-read the manuscript to minimize typographical, grammatical, and bibliographical errors.

Here below is our description on revision according to the reviewer's comments.

Part A (Reviewer 1)

The reviewer's comment: .The tables lack the definition of the acronym FBG.

The authors' Answer: We have explained the definition of the acronym FBS in Tables 1, 2 and 3

All authors revised manuscript according to the steps for submitting revised manuscript.

We found the 56 reference mistakenly turned into 67 using the reference auto-analyser , but we don't how to revise it .

Thank you and the reviewer for the kind advice.

Sincerely yours

Danhong Yang

Corresponding author:

Danhong Yang, MD, PhD, Department of Infectious Diseases Zhejiang Provincial People's Hospital, Hangzhou ,Zhejiang ,310014, People's Republic of China .

Telephone: +86-0571-85893602;

mobile: 136-7581-8175

E-mail: ydh-11@163.com

March 25,2021